Author's response to reviews

Title: Imparting carrier status results detected by universal newborn screening for sickle cell and cystic fibrosis in England: a qualitative study of current practice and policy challenges

Authors:

Hilda Parker (Hilda.Parker@nottingham.ac.uk)
Nadeem Qureshi (Nadeem.Qureshi@nottingham.ac.uk)
Fiona Ulph (Fiona.Ulph@nottingham.ac.uk)
Joe Kai (Joe.Kai@nottingham.ac.uk)

Version: 5 Date: 23 October 2007

Author's response to reviews:

Revisions made:

1. United Kingdom changed to UK and country after post code
2. Competing interests: changes made as requested
3. Footnote removed, now part of manuscript text
4. Funding heading removed and details part of 'acknowledgement'
5. No websites in text